Author's response to reviews

Title: Disparities in lipid management for African Americans and Caucasians with coronary artery disease: A national cross-sectional study

Authors:

Mark W Massing (mmassing@mrnc.org)
Kathleen A Foley (Kathleen_Foley@merck.com)
Lori Carter-Edwards (loriedwards@direcway.com)
Carla A Sueta (Carla_sueta@med.unc.edu)
Charles M Alexander (charles_alexander@merck.com)
Ross J Simpson Jr. (rsimpson@med.unc.edu)

Version: 4 Date: 17 August 2004

Author's response to reviews:

Abstract shortened to < 350 words as requested by the editor.